Massachusetts Institute of Technology Handout 15 6.915: Network and Computer Security October 12, 1995 Professor Ronald L. Rivest

Readings for Lecture 10

Attached are the following papers:

ffl Stalking the Wily Hacker by C. Stoll [3] ffl A Survey of Intrustion-Detection Techniques by T. Lunt [2] ffl An Intrustion-Detection Model by Denning by D. Denning [1] ffl Reflections on Trusting Trust by K. Thompson [4]

References

[1] D. Denning. An intrusion detection model. In IEEE Security and Privacy, 1986. [2] T.F. Lunt. A survey of intrusion detectiion techniques. Computers and Security, 12:405– 418, 1993.

[3] C. Stoll. Stalking the wily hacker. Communications of the ACM, 31(5), 1988. [4] K. Thompson. Reflections on trusting trust. Communications of the ACM, 27(8), 1984.